Author's response to reviews

Title: Protocol for the Development and Testing of an Implementation Strategy for a Complex Housing Intervention

Authors:

Dennis P Watson (dpwatson@iu.edu)
Jeani Young (jcyoung7@iupui.edu)
Emily Q Ahonen (eqahonen@iu.edu)
Huiping Xu (huipxu@iu.edu)
Macey Henderson (macthomp@iu.edu)
Velery Shuman (vshuman@heartlandalliance.org)
Randi Tolliver (rtolliver@heartlandalliance.org)

Version: 2
Date: 5 September 2014

Author's response to reviews:

Dear Editorial Board,

We have changed the title of the manuscript per the Deputy Editor's request. The suggested title change was "Development and Testing of an Implementation Strategy for a Complex Housing Intervention: protocol for a mixed methods program/evaluation". My co-authors and I have changed the title to "Development and Testing of an Implementation Strategy for a Complex Housing Intervention: protocol for a mixed methods study", as we felt the term "evaluation" did not encompass the entire project.

Thank you for your consideration. We look forward to hearing from you.

Dennis Watson